Author’s response to reviews

Title: Patterns of Recurrence after Surgery and Efficacy of Salvage Therapy after Recurrence in Patients with Thoracic Esophageal Squamous Cell Carcinoma

Authors:

Wenjie Ni (niwenjie_1990@163.com)
Jinsong Yang (ray63@126.com)
Wei Deng (sherrydw@126.com)
Zefen Xiao (xiaozefen@sina.com;xiaozefen_bmc@163.com)
Zongmei Zhou (zhouzongmei2013@163.com)
Hongxing Zhang (hongxing733@hotmail.com)
Dongfu Chen (cdf1825@sina.com)
Qinfu Feng (13811300221@163.com)
Jun Liang (liang23400@163.com)
Jima Lv (lvjima419@yahoo.com)
Xiaozhen Wang (wangxiao2008@yahoo.com)
Xin Wang (beryl_wx2000@163.com)
Tao Zhang (zhangt10@126.com)
Nan Bi (binan_email@163.com)
Lei Deng (dengleipumc@163.com)
Wenqing Wang (wwq4350524@hotmail.com)

Version: 2 Date: 05 Feb 2020

Author’s response to reviews:

Editor Comments: Thanks for your comments, here are all my responses.
1. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: I have describe the role of the funding bodies. (Funding section, line 301-302, page 13)

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional suppleinmentary files are cited within the text.

Response: I am sure the manuscript is the final and clean version. All figures are uploaded as separate files and cited within the text.